ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

de Carvalho, Larissa Pinon et al.

Article

Predictors of digital competence of public university employees and the impact on innovative work behavior

Administrative Sciences

Provided in Cooperation with: MDPI – Multidisciplinary Digital Publishing Institute, Basel

Suggested Citation: de Carvalho, Larissa Pinon et al. (2023) : Predictors of digital competence of public university employees and the impact on innovative work behavior, Administrative Sciences, ISSN 2076-3387, MDPI, Basel, Vol. 13, Iss. 5, pp. 1-25, https://doi.org/10.3390/admsci13050131

This Version is available at: https://hdl.handle.net/10419/275596

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



WWW.ECONSTOR.EU

https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.







Article Predictors of Digital Competence of Public University Employees and the Impact on Innovative Work Behavior

Larissa Pinon de Carvalho ¹, Thiago Poleto ^{1,}*⁰, Camila Carvalho Ramos ¹, Fernando de Assis Rodrigues ², Victor Diogho Heuer de Carvalho ³ and Thyago Celso Cavalcante Nepomuceno ⁴

- Graduate Program in Administration, Applied Social Sciences Institute, Federal University of Pará, Belém 66075-110, Brazil
- ² Graduate Program in Information Science, Applied Social Sciences Institute, Federal University of Pará, Belém 66075-110, Brazil
- ³ Sertão Campus, Federal University of Alagoas, Delmiro Gouveia 57480-000, Brazil
- ⁴ Technology Nucleus, Federal University of Pernambuco, Caruaru 55014-900, Brazil
- * Correspondence: thiagopoleto@ufpa.br

Abstract: Digital competence plays an important role in higher education. The literature highlights the adoption and use of digital competence for the development of educational services in Higher Education Institutions (HEIs), but it is still necessary to understand its influence on the innovative behavior of the public official. This study aims to investigate the influence of digital competence, based on the predictors' infrastructure, integration, and digital management, on innovative work behaviors. The research was applied to 540 public employees of a Federal Institution of Higher Education (IFES) in northern Brazil. The response rate for this study was 33.5%. The Structural Equation Modeling approach by Partial Least Squares (PLS-SEM) was used. Empirical findings confirm that public employees with a high level of perception of digital competence more often develop challenging work to generate new innovative behaviors at work. This research was limited to investigating the influence of digital competence on innovative work behaviors of public employees of a Brazilian IFES. Future studies may address other contextual factors in this relationship. One of the practical implications is the need for managers in the education sector to support the construction of guidelines for educational and technological innovation to expand innovative behavior at work.

Keywords: digital competence; innovative work behavior; education innovative; public administration

1. Introduction

Brazilian Higher Education Institutions (HEIs) are challenged to develop innovative educational solutions to meet the requirements of developing new competencies. In this sense, digital technologies contribute to transforming the way public educational services add value to the formation of a new profile of university students (Hashim et al. 2022). The commercial value of digitization does not only involve the integration of new digital technologies into today's infrastructure but also represents how these new technologies can be used to transform business processes and create business value (Castioni et al. 2021). Higher Education Institutions discuss the implementation of concepts such as big data, data vigilance, and artificial intelligence associated with Industry 5.0 (Baig et al. 2020).

In the educational sphere, a large volume of data is produced through online courses and teaching and learning activities. An important factor in educational practices is considered the influence of digitization, algorithm-based media, and machine learning as they create new improvements in and explore the relationships between process digitalization and learning practices, which modify the way people access information and communicate and coordinate services (Van de Wetering et al. 2021; Verhoef et al. 2021). The information and services quality and the complexity of the used systems are also highlighted factors in the digitization process, supporting the automation of academic life (Dospinescu and



Citation: Carvalho, Larissa Pinon de, Thiago Poleto, Camila Carvalho Ramos, Fernando de Assis Rodrigues, Victor Diogho Heuer de Carvalho, and Thyago Celso Cavalcante Nepomuceno. 2023. Predictors of Digital Competence of Public University Employees and the Impact on Innovative Work Behavior. *Administrative Sciences* 13: 131. https://doi.org/10.3390/ admsci13050131

Received: 1 March 2023 Revised: 25 April 2023 Accepted: 26 April 2023 Published: 11 May 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Dospinescu 2020). The digitization of academic life implies finding the surplus of the teaching-learning process with digital tools and increasing the motivation for concrete, effective, and subject-oriented examples, all guided by experienced professionals, such as teachers (Amhag et al. 2019).

Digital technologies integrated with artificial intelligence are necessary to advance flexible educational practices. In recent years, digital technologies have been important during the COVID-19 pandemic, and thoughts about the impact of the pandemic have brought about new learning (Galanti et al. 2023; Hoti et al. 2022). Digital technologies have provided public employees with academic support tasks at home, supported by IT infrastructure resources, process digitization, and management monitoring (Nikou and Aavakare 2021). From this, it is appropriate to prepare managers, teachers, and students for future uncertainties by assuming digital competences; this, in turn, refers to access, knowledge, pro-activity, and mastery of digital technologies with progressive levels of autonomy and learning (Ala-mutka 2011). In this digital context, several factors are known to affect digital teaching competence; these are macro-environmental factors, where the state interferes in technological progress with policies for the development of digital competence for the population; environmental factors, where the university includes digital tools in education to ensure the professional development of teachers through the availability and quality of digital resources; micro-environmental factors, referring to support in the development of digital competence; and personal factors, where the individual feels motivated to aid in the development of digital competence (Pesha 2022).

Digital competence has recently gained a strong prominence in the educational context, being one of the key competencies that citizens in general, and teachers specifically, should master in the society of the future (Cabero-Almenara et al. 2020; Tejada Fernández and Pozos Pérez 2018). On the one hand, the use of technology has become an everyday occurrence due to advances in the creation and exchange of content on the internet and social networks. On the other hand, the development of many individuals depends, for the most part, on adequate and efficient use of ICTs (Basilotta-Gómez-Pablos et al. 2022). The literature addresses several frameworks that determine teachers' digital competence levels. In the European context, the DigCompEdu (Digital Competence Framework for Educators) framework was developed based on six areas of digital competence teachers must have to promote effective, inclusive, and innovative learning strategies using digital tools (Caena and Redecker 2019).

Other previous studies propose frameworks for assessing the digital competence of university teachers and staff (Ciriza-Mendívil et al. 2022; Dias-Trindade and Ferreira 2020; Nikou and Aavakare 2021), analyzing the level of digital competence of teachers in higher education (Guillén-Gámez and Mayorga-Fernández 2020), identifying the factors that influence the development of competence in teachers (Basantes-Andrade et al. 2020; Cabero-Almenara et al. 2020), and stimulating the development of digital teaching competence (Fadli et al. 2020; Gleason and Manca 2020). Finally, they examine the positive impact of digital competence on organizational agility and performance (Ravichandran 2018; Yu and Moon 2021).

Innovation at work is occurring rapidly in sectors such as the automotive industry and startups. However, the innovation transition is slower in some sectors, including education. In the future, formal educational institutions will likely continue to disseminate relevant knowledge to society. However, if those educational institutions take a long time to adapt, the market tends to replace parts of the formal education models with an alternative and more dynamic model. It is important to note that digital competence does not replace all conventional literacy skills. Indeed, these skills are still critical to success in everyday life and citizen education. However, digital competence offers a valuable extension of these skills, allowing those involved in the transformation process to be part of more cohesive work groups, to access information more quickly, and to share it efficiently, regardless of physical location.

In public education, innovative behavior involves workgroup abilities and collaboration with educators and public servants to develop creative solutions to educational problems and challenges. However, it is paramount to emphasize that innovative behavior is not just the responsibility of educators. Educational institutions are part of the process and must encourage and support innovation by providing resources and tools for educators who want to experiment with new teaching approaches. In addition, educational institutions must create an environment that encourages innovative behavior, valuing creativity and critical thinking. The insights provided by the new technological and educational innovation paradigm could be used by individuals with skills that will demand to build a theoretical model of the hypothetical relationship between digital competence and the IWB of civil servants in the future. As individuals affiliated with educational organizations and using digital resources, employee IWBs are inevitably affected by digital competence.

The main objective of this study is to investigate the influence of practices on digital competence on the behavior of innovative work. We approach this objective with the following question: How does digital competence influence innovative behavior in the work of civil servants in a Higher Education Institution?

Among the goals derived from the main objective, we can list the understanding of innovative behavior at work and the analysis of the impact of digital competence on innovative behavior at work. Following this logic and considering that previous studies do not relate digital competence to innovative behavior at work, this study extends the existing literature on innovative work behavior. Therefore, it can be considered opportune and appropriate to understand individual innovation. The relevance of this research is to contribute to the study of individual differences as a predictor of behaviors that implies innovative behavior at work, which may apply to management practices and policies, especially with the proposition of new innovative educational management models focusing on developing effective assessment strategies, including advanced assessment technologies to provide instant feedback, such as the application of artificial intelligence. Another innovative practice is adopting competency-based learning models, which accentuate developing specific skills and competencies rather than simply memorizing facts. This context requires a more personalized approach to education.

To investigate the relationship proposed in this article, we conducted an empirical study with 181 civil servants, who have experience with Business Process Management (BPM) and digitalization of processes, through the application of online questionnaires between October and November 2022; these civil servants were all employed at a Public Institution of Higher Education in Northern Brazil. For data analysis, the Partial Least Squares Structural Equation Modeling Technique (PLS-SEM) was applied using the SmartPLS 4 software package.

The remainder of this article is divided as follows: Section 2 provides the background based on the literature and also defines the hypotheses to be tested; Section 3 presents the obtained results; Section 4 presents the description of the methodological procedures applied in the research; Section 5 makes the necessary comments and brings the theoretical and practical implications of the research and also presents the limitations of the research and indicates directions for future developments. Finally, Section 6 contains the article's conclusions.

2. Literature Background and Hypotheses

Digital competence has recently been a factor of success and organizational competitiveness (Pesha 2022). This competence contains three central factors: digital infrastructure, digital integration, and digital management (Yu and Moon 2021). Digital infrastructure refers to the organization's use of digital-related architecture to adapt organizational processes, specifying structures and responsibilities that meet technological changes. Digital integration refers to the ability to integrate strategies aligned with digital technologies. Digital management relates to the management of digital-related competencies and experiences by the organization, to meet the requirements of infrastructure support, and digital training of employees (Kindermann et al. 2021; Yu and Moon 2021).

Innovative Work Behavior (IWB) refers to the employee's behavior aimed at generating, introducing, and/or applying (within a function, group, or organization) new ideas, processes, products, or procedures intended to benefit the unit of work (De Spiegelaere et al. 2014). The concept implies that individuals manage, promote, and realize innovative ideas for improvements in, for example, products or processes. The IWB refers to the exploration, generation, defense, and implementation of ideas (De Jong and Den Hartog 2010). Prior research supports innovative work behavior. First, for young start-ups, authentic leadership has a significant effect on innovative work behavior among indian employees (Sengupta et al. 2023). Second, challenge stressors improve innovative work behavior (Yu et al. 2023). In this context, new scales for innovative work behavior (IWB) were presented for the success of educational institutions (Ayoub et al. 2023).

2.1. Digital Competence: Digital Infrastructure and Innovative Work Behavior

New educational contexts require digital technologies, such as video conferencing platforms, cloud file storage platforms, messaging applications, and management information systems. They allow employees to spend most of their time connected, which is an opportunity for educational improvement and innovation as technologies involve changes in educational practices and processes (Ciriza-Mendívil et al. 2022; Rivero and Mur 2015). First, digital technologies provide massive access to data, information, and resources. Employees can search for information, learn new competences, and learn new technologies and trends much more easily than before. This can drive creativity and innovation, as employees have more access to information and resources to help them create new ideas and ways to improve work processes.

Second, they facilitate communication and collaboration because they allow employees to connect and share information easily, regardless of geographic distance (Hoti et al. 2022; Li et al. 2009), which can induce innovative ideas. Third, they promote flexibility and mobility to employees, allowing them to work from anywhere and at any time (Sengupta and Al-Khalifa 2022). This may create a more innovation-friendly environment as employees may have more time and space to think and explore new ideas. Finally, digital infrastructure could also increase the efficiency of work processes (Saeed 2019), providing more time for innovation. For example, automation tools could help to eliminate repetitive tasks and allow employees to focus on more creative activities to stimulate innovative behavior at work.

Authors have pointed out that an organizational climate favorable to innovation influences innovative employee behaviors (Afsar and Umrani 2020). Similarly, confidence in innovation influences a high level of innovative work behavior. Confidence facilitates the desire among employees to contribute new insights, as it results in an open-minded atmosphere in which employees feel confident in bringing suggestions and contributions to the discussion (Afsar et al. 2015). In addition, studies indicate that a supportive environment, both of management and the work team, arouses individuals' interest in proposing innovative ideas, which consequently impacts positively and directly on innovative work behavior (Attiq et al. 2017). In addition, individuals with motivation to learn and learner guidance tend to see new and difficult tasks, such as IWB, as both challenging and as opportunities to learn (Afsar and Umrani 2020; Atitumpong and Badir 2018).

Employees will not be replaced, but technology is available to add to their work in order to establish new ways to integrate these technologies into their processes (Antonucci et al. 2021). Its functions will be performed more effectively, without wasting hours finding information or occupying itself with paper stacks. Faced with this scenario, technology provides more assertive demands, greater flexibility, and access to information; this changes employees' work behavior because it allows them to be resilient, adaptive, and directed to collective alignment. Moreover, one of the reasons technology has advanced in recent decades is that digital transformation has expanded the human capacity to reflect on routine

issues and challenges and find creative and innovative solutions. Therefore, it is interesting to test to what extent digital infrastructure can stimulate innovative work behavior. On this basis, hypothesis 1 (H1) is suggested:

Hypothesis 1. *Digital infrastructure is positively related to Innovative Work Behavior.*

2.2. Digital Competence: Digital Integration and Innovative Work Behavior

Digital technologies have an important role to play in public educational institutions (Hashim et al. 2022). They can be used to increase the quality of teaching and learning through the use of distance learning tools, learning platforms, and class management, as well as making the management and administration of institutions more efficient. This is achieved through the use of management information systems, which enable the integration of data between various sectors, to allow the sharing and free use of information from other units in real-time, which can help employees get to know the team, understand team processes, and have a broader view of the work being done, and to contribute to more efficient teamwork, impacting on the speed of procedural flows, meaning that less time is required to complete a task (Ciriza-Mendívil et al. 2022; Hoti et al. 2022).

To integrate digital technologies into organizations, it is necessary to refit and change the work processes, to facilitate new processes with new proposals of managerial and organizational alignment, and to use new ways of integrating people (Antonucci et al. 2021; Kindermann et al. 2021; Quinton et al. 2018). For this, it is essential that the institution's strategic planning includes an effective digital strategy (Bharadwaj et al. 2013). This requires adequate infrastructure, employee training, clear and well-defined usage policies, and adequate investments (Sengupta and Al-Khalifa 2022).

In this sense, digital technology could be critical to organizations in supporting creative and innovative processes. It allows employees easy access to information and collaboration, which may lead them to be more innovative and explore new possibilities (Antonucci et al. 2021; Austin et al. 2012). This is especially useful when employees request to gather information from various sources in order to make decisions or resolve issues quickly, or when they need to exchange information with others quickly and efficiently. In this way, organizations must find ways to innovate through technologies, developing digital strategies that drive innovative behaviors (Hess et al. 2016).

Self-esteem, self-efficacy, and autonomy at work could be factors that influence the development of innovative behaviors at work (Atitumpong and Badir 2018; Attiq et al. 2017; Bos-Nehles and Veenendaal 2019). In addition, job crafting, a process by which employees act as active agents, shaping and redesigning their work processes to ensure a good person-to-work fit in their work environment, facilitates the process of creating changes, thereby inducing innovative behaviors (Afsar et al. 2019).

In a scenario where technology is present full-time in organizations, it is necessary to understand the interactive potential of the electronic tools used day-to-day and to obtain the ability to use virtual resources that allow the extraction of information, intelligent connections and communication with different sectors of the organization; this induces employees to reflect on innovative solutions to improve the digital integration of organization. Therefore, this research investigates whether digital integration in organizations has a positive impact on innovative behavior at work. On this basis, hypothesis 2 (H2) is presented:

Hypothesis 2. *Digital integration is positively related to Innovative Work Behavior.*

2.3. Digital Competence: Digital Management and Innovative Work Behavior

Due to the intensifying demand for digital competence in the labor market, employers are interested in developing the digital competence of future experts as part of their training at university (Pesha 2022). Therefore, the development of teachers' digital competence is a relevant issue, since the construction of students' ability to live in a digital environment

depends, to a large extent, on their digital literacy. The use of digital technologies in the implementation of curricula increases the level of assimilation of course material by students and helps in the formation of digital competence of future specialists (Attali and Arieli-attali 2015). Thus, it is appropriate that educational institutions develop digital competence, both among students and between teachers and university technicians, to increase efficiency in teaching-learning and administrative processes. To do this, institutions must conduct consistent digital management through conducting training to develop digital knowledge and competence (Galanti et al. 2023).

In this context, the ability to manage and analyze information, as well as communication competence, learning autonomy, and critical thinking, have become common characteristics of professionals in the digitalized information age, which induces individuals to creativity in the elaboration of strategies and problem-solving in an environment of digital solutions. The digital scenario encourages employees to develop new competence and expand existing expertise in order to obtain the ability to constantly resignify and adapt to the evolution of the modern world, inducing them into behaviors focused on innovation (Galanti et al. 2023).

When managing digital knowledge, managers should adopt a transformational leadership style that inspires employees individually by developing a strong sense of shared vision and belonging to the organization; this can encourage employees to engage in innovative work behaviors (Afsar and Umrani 2020; Afsar et al. 2014; Afsar et al. 2019; Maria Stock et al. 2017). In addition, the sharing of digital knowledge nourishes employees with enough information to generate and implement new ideas, and to provide innovative work behaviors (Afsar et al. 2019; Bos-Nehles and Veenendaal 2019). Furthermore, HR practices, such as digital-focused training and development, rewards for employees who stand out in the digital industry, and feedback related to digital practices, can positively influence innovative behaviors at work (Afsar et al. 2019; Bos-Nehles et al. 2017).

By investing in the development of knowledge and competence necessary to leverage the use of technology, organizations improve the digital experience of employees and, as a result, increase the innovation capacity and creativity of employees, making them want to engage, increasingly, in the evolution of the organization through innovative ideas and solutions. For this reason, this research examines the influence of digital management on innovative work behaviors. On this basis, hypothesis 3 (H3) is presented:

Hypothesis 3. Digital management is positively related to Innovative Work Behavior.

Figure 1 describes the central conceptual research model of this study in order to present the relationship between digital competence and innovative behavior at work. First, the relationship between digital infrastructure and innovative behavior at work was explored. Second, the relationship between digital integration and innovative behavior at work was evaluated. Finally, the relationship between digital management and innovative behavior at work was evaluated.



Figure 1. Research Model Digital Competence and Innovative Work Behavior.

2.4. The Negative Effects of Digital Competence over Innovative Work Behavior

Despite the positive impacts of the digital competence elements (infrastructure, integration, and management) on innovative work behavior, there are also elements that act in an inverse way, causing negative impacts; these include physical and mental discomfort and stress, digital distractions, motivation for long periods of activity, learning difficulties, etc. In terms of discomfort, this may be related to receiving direct instructions to perform the work in the traditional (presential) way (Darazha et al. 2021), causing, for instance, anxiety (Gansser and Schultz 2020). Burnout syndrome, for instance, is a potentially negative effect, causing a combination of symptoms including physical, emotional, and social stress, thereby decreasing work performance (Ramos et al. 2023).

Digital distraction is another problem that can emerge when developing people's digital competences, since in the cyber world several distractive elements can absorb the workers' attention and even cause work interruptions and a decrease in productivity (Gerbaulet and Korn 2018). Despite being a tool in the digital transformation, integration, and communication, social media can also be a harm for the worker's attention, directly affecting his performance in developing work-related tasks, as it is easy to shift from the digital working area into digital social networks and other related channels (Andersson et al. 2021).

Motivational barriers can also appear in the three hypothesis relations of interest, especially from a psychological perspective, since the users may not feel able to use the digital tools. This can occur as a result of their socioeconomic level or personal learning difficulties, which may make it difficult to absorb the necessary experience needed to use these tools productively (Dobson et al. 2022). There is another perspective for these difficulties in adopting digital work and competences; cognitive difficulties, related to neurological disorders, can disable workers. These include Parkinson's disease, different types of sclerosis, or even people who have suffered strokes and are undergoing rehabilitation processes (Lindberg et al. 2021).

Independent of the case, each potential cause of negative effects in the relations of digital transformation should be considered in the counter hypothesis to those previously de-fined, if not by direct testing, then at least by the effects it presents.

3. Results

3.1. Measurement Model Test

The objective of the Measurement Model is to evaluate the quality of measurement of latent variables that were used to test the hypotheses established for this research. Thus, reliability, convergent validity, and discriminant validity of each construct were evaluated.

Figure 2 shows the Measurement Model test. This test shows the factor loadings for each item, which range from 0.608 to 0.852. In addition, it presents Cronbach's Alpha for each construct, ranging from 0.703 to 0.788, and structural coefficients from 0.208 to 0.394.



Figure 2. Measurement Model Test.

In terms of convergent validity, the AVE (Average Variance Extracted) score of each construct must exceed 0.50 (Fornell and Larcker 1981). According to Table 1, the AVE scores of all constructs were >0.5, providing sufficient evidence of convergent validity after discarding the item "DI4" of the Digital Infrastructure (DI) construct and the items "DIT1" and "DIT2" of the Digital Integration (DIT) construct; this is because their low factor loadings (<0.40) were decreasing the AVEs of these constructs (Fornell and Larcker 1981; Henseler et al. 2009; Ringle et al. 2014).

Table 1. Loads, Cronbach's Alpha, composite reliability and AVE.

Construct	Items	Loads	Cronbach's Alpha	Composite Reliability (rho_a)	Composite Reliability (rho_c)	AVE
	DI1	0.760				
Digital Infrastructure (DI)	DI2	0.729	0 702	0 702	0.01/	0.50
Digital Illiastructure (DI)	DI3	0.744	0.703	0.703	0.816	0.526
	DI5	0.664				
Digital Integration (DIT)	DIT3	0.755				
	DIT4	0.811	0.747	0.810	0.848	0.651
	DIT5	0.852				
	DM1	0.777		0.784	0.847	
	DM2	0.676				
Digital Management (DM)	DM3	0.765	0.776			0.527
	DM4	0.683				
	DM5	0.722				
	IWB1	0.786				
	IWB2	0.787				
Innovative Work Behavior (IWB)	IWB3	0.608	0.788	0.802	0.856	0.548
	IWB4	0.851				
	IWB5	0.638				

In terms of reliability (internal consistency), Cronbach's α and composite reliability (CR) values should exceed 0.7 (Hair et al. 2017). According to Table 1, all Cronbach's α values and composite reliability (CR) were >0.7, suggesting good internal consistency. In terms of indicator reliability, the factorial loading of each indicator must be greater than 0.7. Commonly, the indicators with loadings between 0.40 and 0.70 should only be removed from the scale if their exclusion leads to an increase in the composite reliability above the value suggested limit (>0.7) (Hair et al. 2017). According to Table 1, more than 75% of the indicator loads were > 0.7, but 4 loads (DI5 = 0.664; DM2 = 0.676; DM4 = 0.683; IWB5 = 0.638) are in the range between 0.4 and 0.7. When examining the composite reliability of all constructs, without excluding the factor loadings >0.4 and <0.7, it appears that the CRs already had a value > 0.7. By excluding them, it was found that the composite reliability remained with a load greater than 0.7, so the exclusion of items was not necessary, suggesting good reliability (Hair et al. 2017).

Three criteria were analyzed for discriminant validity: (1) Fornell–Larcker; (2) Crossloading and (3) HTMT (The Heterotrait-monotrait ratio). Fornell-Larcker describes that a latent construct shares more variance with its assigned indicators when compared to another latent variable in the structural model. The AVE of each latent construct should be greater than the highest square correlation of the latent construct with any other latent construct (Fornell and Larcker 1981; Hair et al. 2017). Table 2 presents the results and indicates the fulfillment of the assumptions of the Fornell-Larcker. Cross-loading compares the loading of an item on its related construct with its cross-loading on other constructs (Liu et al. 2018). According to Table 3, the external loadings of the indicators of the constructs themselves were all higher when compared to the cross-loadings. The Heterotrait-monotrait ratio (HTMT) is the mean of all correlations of indicators across constructs measuring different constructs (the heterotrait-heteromethod correlations) relative to the geometric mean of the average correlations of indicators measuring the same construct (the monotraitheteromethod correlations). O HTMT must be less than 0.90 (Henseler et al. 2015). As shown in Table 4, all correlations comply with the criterion. Therefore, according to the Fornell-Larcker criterion, Cross-loading, and the Heterotrait-monotrait ratio, all constructs have good discriminant validity.

	DI	DIT	DM	IWB
DI	0.725			
DIT	0.389	0.807		
DM	0.574	0.509	0.726	
IWB	0.549	0.503	0.639	0.740

Table 2. Discriminated validity test (Fornell-Larcker criterion).

DI = Digital Infrastructure; DIT = Digital Integration; DM = Digital Management; IWB = Innovative Work Behavior.

	Digital Infrastructure	Digital Integration	Digital Management	Innovative Work Behavior
DI1	0.760	0.368	0.427	0.460
DI2	0.729	0.104	0.345	0.287
DI3	0.744	0.317	0.345	0.380
DI5	0.664	0.277	0.515	0.420
DIT3	0.207	0.755	0.341	0.309
DIT4	0.270	0.811	0.319	0.318
DIT5	0.410	0.852	0.515	0.524
DM1	0.433	0.401	0.777	0.553
DM2	0.371	0.331	0.676	0.371
DM3	0.484	0.359	0.765	0.439
DM4	0.424	0.361	0.683	0.498
DM5	0.361	0.387	0.722	0.419

 Table 3. Discriminated validity test (Cross loadings).

	Digital Infrastructure	Digital Integration	Digital Management	Innovative Work Behavior
IWB1	0.369	0.378	0.451	0.786
IWB2	0.508	0.388	0.483	0.787
IWB3	0.263	0.267	0.343	0.608
IWB4	0.447	0.442	0.499	0.851
IWB5	0.396	0.355	0.549	0.638

Table 3. Cont.

DI = Digital Infrastructure; DIT = Digital Integration; DM = Digital Management; IWB = Innovative Work Behavior.

Table 4. Discriminated validity test (Heterotrait-monotrait ratio—HTMT).

	Digital Infrastructure	Digital Integration	Digital Management	Innovative Work Behavior
Digital Infrastructure			0.755	0.702
Digital Integration	0.470		0.630	0.607
Digital Management				0.795
Innovative Work Behavior				

The model was tested to ensure that collinearity was not a problem; to ensure this, variance inflation factors (VIFs) must be less than 3.3 (Kock 2015). As shown in Table 5, all constructs had VIF below the conservative threshold of 3.3, with a minimum value of 1.375 and a maximum value of 1.740, thus suggesting that multicollinearity was not a critical problem in the investigation.

Table 5. Colinearity statistics (VIF)-Inner model.

Path	VIF
Digital Infrastructure -> Innovative Work Behavior	1.520
Digital Integration -> Innovative Work Behavior	1.375
Digital Management -> Innovative Work Behavior	1.740

3.2. Structural Model Test

The next step was to evaluate the results of the structural model and test the proposed hypotheses. The quality of the structural model was evaluated using two indicators. The first indicator evaluates the explanatory power of the model through the value of R^2 , R^2 , and the value of the explanatory effect f^2 . The second indicator evaluates the predictive ability of the model by analyzing the significance of the path coefficients, the predictive correlation Q^2 , and the value of the effect q^2 .

 R^2 reflects the proportion of variation endogenous constructs that can be explained by the exogenous constructs in the model (Hair et al. 2017). According to Table 6, the three exogenous constructs together (DI, DIT e DM) explain 48.9% of the variance of the endogenous construct IWB ($R^2 = 0.489$; R^2 adjusted = 0.481), which shows a moderate explanatory power (Hair et al. 2017). The value of the explanatory effect f^2 measures the amount of change in value R^2 after excluding the specific exogenous variables in the model (Hair et al. 2017).

The analysis of the statistical significance of the path coefficients was calculated using the bootstrap with five thousand samples for a one-tailed test with a significance level of 0.05 (Hair et al. 2017). For the choice of five thousand bootstrap samples, an evaluation was performed. First, a smaller number of bootstrap sub-samples were chosen (for example, 1000) to be drawn randomly and estimated with the PLS-SEM algorithm because of the shorter processing time. Subsequently, the preparation of the final results was changed to a large number of subsamples bootstrap (for example, 10,000). Thus, the modification aims to ensure the stability of the results. Understanding that five thousand samples are sufficient for the research, the same samples were applied. Furthermore, the one-tailed test was chosen as positive assumptions were made in the research hypotheses (Kock 2015). According to the value assessment criterion f^2 of Cohen, as seen in Table 6, all exogenous variables have significant explanatory power in the model. The explanatory power of the exogenous variable "DM" is considered higher ($f^2 = 0.175$ —average effect) when compared to the explanatory power of the exogenous variables "DI" and "DIT" ($f^2 = 0.076$ and $f^2 = 0.062$ —small effect) (Cohen 1988).

Table 6. Model's explanatory capacity $(R^2, R^2 \text{ adjusted}, f^2)$.

Endogenous Latent Variable	<i>R</i> ²	R ² Adjusted	Effect Size <i>f</i> ²
Innovative Work Behavior (IWB)	0.489	0.481	
Path			
Digital Infrastructure (DI) -> Innovative Work Behavior (IWB)			0.076
Digital Integration (DIT) -> Innovative Work Behavior (IWB)			0.062
Digital Management (DM) -> Innovative Work Behavior (IWB)			0.175

Through the bootstrap procedure, we identified that the path coefficients DI -> IWB, DIT -> IWB, and DM -> IWB are significant (p < 0.01), as shown in Figure 3. In addition, a predictive correlation was analyzed (Q^2 of Stone–Geisser) that represents an evaluation criterion for the predictive relevance of cross-validity of the path model PLS and aims to evaluate the accuracy of the fitted model and should be not equal to null (Hair et al. 2017). According to Table 7, Q^2 has a value of 0.451, indicating that the exogenous constructs have predictive relevance for the endogenous construct under consideration. In addition, the relative impact of predictive relevance was compared by measuring the effect size q^2 , which allows us to evaluate the contribution of an exogenous construct to the value Q^2 of an endogenous latent variable. According to Table 7, it can be seen that the exogenous construct "DM" has greater predictive relevance ($q^2 = 0.162$ —average effect), when compared to the predictive relevance of "DI" "DIT" ($q^2 = 0.035$ e $q^2 = 0.038$ —small effect).



Figure 3. Structural Model Test.

Constructs	02	Ω^2 Excluded	Effect Size a^2
Constructs	Q	Q Excluded	Effect Size q
Innovative Work Behavior (IWB)	0.451		
Digital Infrastructure (DI)		0.421	0.055
Digital Integration (DIT)		0.430	0.038
Digital Management (DM)		0.362	0.162

Table 7. Model's Predictive Capacity (Q^2 and effect size q^2).

According to Figure 3 and Table 8, the structural model confirms the positive associations between DI and IWB (β = 0.242, p < 0.001), DIT and IWB (β = 0.208, p < 0.01), and between DM and IWB (β = 0.394, p < 0.001). Therefore, the path coefficients referring to the three presented hypotheses are significant. Thus, structural model supports all the hypotheses H1, H2, and H3. The exogenous construct DM exerts a stronger effect on the endogenous construct IWB, followed by DI and DIT, respectively. Also, according to the T statistics, the intensity of the DM and IWB relationship is more prominent (Hair et al. 2017; Nascimento and Macedo 2016).

Table 8. Hypothesis testing results.

Hypothesis	Path	Original Sample (O)	Sample Mean (M)	Standard Deviation (STDEV)	T Statistics (O/STDEV)	p Values	Result
H1 (+)	DI -> IWB	0.242	0.247	0.064	3.782	0.000	Supported
H2 (+)	DIT -> IWB	0.208	0.209	0.081	2.574	0.005	Supported
H3 (+)	DM -> IWB	0.394	0.397	0.076	5.189	0.000	Supported

Furthermore, a 5-point Likert scale was used to understand the behavior of the constructs through the descriptive statistics of the questions that compose them. According to Table 9, the average, median, minimum, maximum, and standard deviation of each question are verified. Regarding the Digital Infrastructure (DI) construct, all questions (DI1, DI2, DI3, and DI5) are close to scale 4. Regarding the Digital Integration construct (DIT), one question (DIT3) is close to scale 4, and two questions (DIT4 and DIT5) are close to scale 3. Regarding the construct of Digital Management (DM), all questions (DM1, DM2, DM3, DM4, and DM5) are close to scale 4. Finally, in the construct Innovative Work Behavior (IWB), all questions (IWB1, IWB2, IWB3, IWB4, and IWB5) are close to scale 4. Additionally, when analyzing Table 10, it is possible to observe that the four constructs (Digital Infrastructure, Digital Integration, Digital Management, and Behavior of Innovative Work) had an average of around 4 (DI = 4.328; DIT = 3.569; DM = 4.230; IWB = 4.052) and the median of all constructs were close to the means (DI = 4.750; DIT = 3.667; DM = 4.400; IWB = 4.200). None of the constructs obtained averages lower than the midpoints of the scales.

Table 9. Descriptive statistics of items for each construct.

	Mean	Median	Observed Min	Observed Max	Standard Deviation
DI1	4.365	5.000	2.000	5.000	0.873
DI2	4.459	5.000	1.000	5.000	0.837
DI3	4.343	5.000	2.000	5.000	0.789
DI5	4.144	4.000	2.000	5.000	0.766
DIT3	3.917	4.000	1.000	5.000	1.092
DIT4	3.365	3.000	1.000	5.000	1.239
DIT5	3.425	4.000	1.000	5.000	1.296

	Mean	Median	Observed Min	Observed Max	Standard Deviation
DM1	4.011	4.000	1.000	5.000	0.975
DM2	4.227	4.000	2.000	5.000	0.779
DM3	4.138	4.000	1.000	5.000	0.799
DM4	4.376	5.000	2.000	5.000	0.816
DM5	4.398	5.000	1.000	5.000	0.791
IWB1	4.309	4.000	1.000	5.000	0.837
IWB2	3.972	4.000	1.000	5.000	0.966
IWB3	3.646	4.000	1.000	5.000	1.150
IWB4	4.055	4.000	2.000	5.000	0.973
IWB5	4.276	5.000	1.000	5.000	0.842
11100	1.270	0.000	1.000	0.000	0.012

Table 9. Cont.

DI = Digital Infrastructure; DIT = Digital Integration; DM = Digital Management; IWB = Innovative Work Behavior.

Table 10. Descriptive statistics of the constructs.

	Mean	Median
DI	4.328	4.750
DIT	3.569	3.667
DM	4.230	4.400
IWB	4.052	4.200

DI = Digital Infrastructure; DIT = Digital Integration; DM = Digital Management; IWB = Innovative Work Behavior.

4. Materials and Methods

4.1. Sample and Procedures

The present study was developed in a Federal Institution of Brazilian Higher Education, recognized for its wide representativeness in the Northern region of Brazil. The criteria for choosing this institution as the object of study was, first, due to the implementation of the Business Process Management (BPM) methodology. Second, the implementation of digital systems in the conduct of academic and administrative processes, including a User Service System, an Integrated Asset, Administration and Contracts System, an Integrated Academic Activities Management System, an Integrated Human Resources Management System and an Integrated Planning and Management System; these systems allow demands to be carried out electronically, generating procedural speed and human or material resource savings.

The conceptual research model was tested in a sample of 540 civil servants, composed of administrative technicians in education and teachers invested in a management function, each of whom already had experience with BPM and the digitization of administrative and/or academic processes. The sample was considered adequate for studies of this nature, considering that the PLS-SEM tool was used; this tool is characterized by being a robust technique that does not require assumptions and has few estimation problems, serving both small and large samples (Hair et al. 2017). The sample was acquired through two channels. First, a survey was carried out in the Process Management Portal of the Institution; this was performed to verify the sectors that have organizational processes already mapped in its repository. After identifying the sectors, it was investigated which work with digital processes; this was achieved through research carried out in the Portal of the Integrated Management System and on the website of the Commission of Electronic Administrative Process for the institution. Finally, the sectors that work with BPM but do not have digital processes were excluded. As such, the final sample of 540 employees, distributed in 12 sectors of the institution, was reached.

Regarding the data collection procedure, a pre-test of the questionnaire was performed to provide improvement in the quality of the items addressed and the correction of writing problems. As becoming a fundamental instrument for the research, this step was vital; it was carried out without great difficulties by the applicators and understood by the respondents, increasing the efficiency and efficacy of the study. The pre-test was applied from 25 to 27 May 2022, with ten specialists in the area (five administrative technicians in education and five teachers invested in a management function). Among the technicians, three presented suggestions for improvement regarding the understanding of the statements and the inclusion of items in the questionnaire; among the teachers, two presented suggestions for improvement regarding the understanding of the statements. After this feedback, the suggestions were integrated into the final version of the questionnaire. Subsequently, a total of 540 questionnaires (Appendix A) were sent between October and November 2022, of which 181 were returned, obtaining a response rate of 33.5%; this is representative considering the target audience, the data collection method, and the time required to answer the statements, and the results provided a solid basis for data analysis and confidence of empirical results. For online surveys, where there was no prior relationship with the recipients, a response rate of 20% to 30% is considered excellent. Participation occurred voluntarily and the study was confidential. Among the respondents, 50.3% were female and 45.3% were between 31 and 40 years old. Regarding the stocking unit, many of the research participants (30.9%) worked at the Institute of Geosciences (IG), followed by the Pro-Rectory of Personnel Development and Management (15.5%). In addition, 44.2% had a specialization/MBA and 32% had a Master's degree; of these, most had a degree in Business Administration (30.4%).

4.2. Measures

In order to measure the relationship between digital competence and innovative work behavior (IWB), a self-administered online questionnaire was constructed. This questionnaire was adapted from previous studies, including Dahiya and Raghuvanshi (2021); Redecker (2017); and Yu and Moon (2021). The questionnaire was organized in two stages. The first related to the constructs to be analyzed; the second related to the identification of the respondent's profile. Table 11 describes the items used to measure the various constructs: Digital Infrastructure, Digital Integration, Digital Management, and Innovative Work Behavior. These were measured based on a 5-point Likert scale. The choice of the Likert Scale for the research is justified because it is easy to understand and accessible to a large number of participants. The Likert Scale allows for more accurate results than other larger or smaller scales, as it provides enough responses to capture the nuance and variation in participants' opinions, is easily statistically analyzed, and makes survey results more reliable and easier to interpret.

Table 11. Definition of constructs.	

Construct	Description	References
Digital Infrastructure (DI)	Refers to the institution's ownership and use of digital-related infrastructure.	(Redecker 2017; Yu and Moon 2021)
Digital Integration (DIT)	Refers to the institution's integration of various digital-related strategies, technologies and knowledge resources.	(Redecker 2017; Yu and Moon 2021)
Digital Management (DM)	Refers to the acquisition of competence and experience related to digital by the management of the institution.	(Redecker 2017; Yu and Moon 2021)
Innovative Work Behavior (IWB)	Behaviors of "individuals" directed to the intentional initiation and introduction of new and useful ideas, processes, products or procedures.	(Dahiya and Raghuvanshi 2021)

4.3. Data Analysis

The present research seeks to analyze and validate the proposed research model. For this, the Technique of Modeling Partial Least Square Structural Equations (PLS-SEM) was used in the SmartPLS 4 software package. The PLS-SEM statistical technique is mainly used to develop theories in exploratory research or to extend existing research by explaining the variance in the dependent variables when examining the model. This technique has been widely adopted in social science research and is used to analyze multiple relationships between unobservable factors (Hair et al. 2017). Therefore, PLS-SEM was considered suitable for this study because the aim of the research is prediction and theory development, making PLS-SEM the most appropriate method. In addition, PLS-SEM facilitates working with a broader range of problems than, for example, the CB-SEM, given its ability to work efficiently with different sample sizes and more complex models; alongside this, its assumptions seem less restrictive about the data and it presents fewer estimation problems. Finally, PLS-SEM allows for the simultaneous estimation of multiple causal relationships between one or more independent variables and one or more dependent variables, and to explore data in search of patterns and relationships.

The analysis was based on two steps. First, the Measurement Model Test was performed, which shows the operationalization of constructs through a set of indicators. Its analysis assesses how much this set of variables represents the construct (Hair et al. 2017). Second, the Structural Model Test was performed, evidencing the constructs' relationships (paths).

5. Discussion, Implications, Limitations, and Future Research Directions

5.1. Discussion

This study investigated the influence of digital competence, through three central factors (infrastructure, integration, and digital management), on innovative work behaviors of public employees of a Brazilian Federal Higher Education Institution. The results indicate that support for digital competence can help the institution to achieve innovative behaviors. In this scenario, infrastructure, integration, and digital management are interdependent factors because they are linked in a continuous cycle; as such, it is up to the university's public management to lead them effectively. Otherwise, access to digital competence may be compromised, which, consequently, will negatively affect the innovative behavior of the individual.

We showed that digital infrastructure (DI) make a positive influence on innovative work behavior (IWB), confirming the first hypothesis. The findings confirm the understanding presented in previous research that states that digitization and the application of information technologies are an opportunity for improvement, as it paves the way for future innovations and research (Ciriza-Mendívil et al. 2022; Paschou et al. 2020). Employees are encouraged to propose innovative ideas that could add value to their demands when encountering technologies that improve their work processes and meet educational demands (Battistelli et al. 2013). Other studies empirically confirm that an environment of confidence in innovation and support from the management and the workforce allows employees to feel comfortable and safe in sharing innovative ideas (Afsar et al. 2015; Attiq et al. 2017).

This paper discusses aspects related to the organizational environment and contributes to knowledge about digital competence. We demonstrate that providing an efficacious digital infrastructure to acquire appropriate technologies for the work environment and necessary operational support for a public institution to generate innovative behaviors is necessary. The absence of a proper infrastructure will make the innovative process unfeasible. Access to digital technologies by public institutions provides employees with a range of possibilities for developing their potential to favor communication and the exchange of information, thereby encouraging greater participation in the construction of knowledge.

Our results indicated that digital integration influences innovative work behavior, confirming the research's second hypothesis. These results are consistent with other recent research that states that digital technology provides new ways to organize organizational activities, builds trust in the actors' ecosystem, and influences innovation (Weking et al. 2020). The readjustment of work processes to make them digital and integrate them into the organization's digital strategy is characterized as a procedure that involves changes, adaptations, and a certain degree of difficulty. Specifically, innovative actions could be valuable, especially when considering that the involvement of employees in the generation

and implementation of new and useful ideas would increase the likelihood of adaptation to changes and overcoming work-related difficulties (Battistelli et al. 2013).

Previous research has found that when an employee has autonomy and self-esteem, and feels confident in developing changes in their work processes, the chances of them engaging in innovative behaviors are higher (Attiq et al. 2017; Battistelli et al. 2013). Other studies claim that job crafting, a process by which employees could redesign processes to ensure a good fit with work, fortifies innovative work behaviors (Afsar et al. 2019). All these IWB determinants can aggregate and strengthen the relationship of digital integration with innovative behaviors at work.

Finally, digital management positively influences innovative work behavior, confirming the third hypothesis. Given the increasingly frequent use of digital technologies, both in teaching-learning processes and in public institutions' administrative processes, public service employees need to pursue training in digital competencies to acquire technological competence and know-how in order to develop both systems use competence and internal management competence to execute a given strategy (Kindermann et al. 2021). For example, Lego sees its employees' digital know-how as a key-factor enabler of its digital strategy (Sawy et al. 2016). Digitally strong organizations strive to acquire individuals to build and renew competence that allows them to digitize value-creation processes and outcomes.

Furthermore, research argues that knowledge sharing creates opportunities for innovative work behaviors (Afsar et al. 2019; Bos-Nehles and Veenendaal 2019). In this way, digital knowledge sharing could become a driving force for innovative behaviors. Other findings show that HR practices, such as training and development, reward, job safety, job demand, and feedback, are positively related to IWB (Bos-Nehles et al. 2017). Moreover, leadership styles, such as transformational leadership and leadership change member-leader, add freedom and autonomy to employees to carry out their work, inducing them to suggest innovative ideas and suggestions in the workplace (Afsar and Umrani 2020; Afsar et al. 2014; Atitumpong and Badir 2018). In addition, to stimulate innovative employee behavior, the public institution should promote training and digital-related rewards to provide leadership that supports employee participation in decisions and encourages innovation in the workplace (Sia et al. 2016).

5.2. Theoretical Implications

From a theoretical perspective, a key contribution of this research is the development of a new theoretical and empirically assessed model of relations between Digital Competence and Innovative Behavior at Work that enriches the literature. According to the researchers' knowledge and available information, this study investigates the perceptions of workers in public education institutions and their internal conditions (digital infrastructure, digital integration, and digital management) that influence innovative behavior at work. This investigation represents an advance in the formulation and conceptualization of the theory of individual behavior. This research confirms that the adoption of new technologies and digital management models are at the heart of advances in innovative behavior at work. Furthermore, the results are consistent with our arguments that organizations that continue to invest in technology and educational management are more inclined to implement innovative behavior actions at work as part of a workforce differentiation strategy.

Understanding the concept of innovative behavior contributes to analyzing specific occupations for innovation activities in research and development and new software acquisitions. On the one hand, this deepening improves the results in the education sector; it contributes to the investments and essential obligations necessary to overcome the impediment for both higher education institutions and public institution workers to invest in specific intermediate-level digital skills. On the other hand, it highlights the importance of widely distributed capabilities in the innovation-focused workforce. These findings contribute to the knowledge of technology-based educational management and innovative behavior, enrich the literature, and discuss, in general, how to agree on their training needs for innovative behavior. The role of public service managers and technicians is considered

to be a participatory vision, where the promotion of the new model of digital competence development is prioritized. This research alerts managers to the quality role of higher education in providing essential training for the development of new capabilities of the higher-level workforce. This research adds value to the management education literature by investigating innovative behavior at work from a developing Latin American country such as Brazil.

5.3. Managerial Implications

The results of this research provide several practical implications. First, educational managers should create practices that enable innovative behavior among employees. Given this scenario, it is recommended that university public management centralize the activities of Technological Innovation to strengthen digital competence and innovative behavior at work, expand the relationship between university-company and society, encourage dissemination and application of knowledge, innovation, and entrepreneurship, and to creating new cooperation, incubation of companies, and encourage technology transfer.

Second, educational managers should develop a new normative framework to be incorporated into the Institutional Development Plan (IDP), with guidelines for educational and technological innovation, and expand innovative behavior at work through the construction of an Integrated People Management Policy. On the other hand, recruitment and selection should adopt new criteria that assess the innovative behavior of the civil servant. The choice of employees to assume management positions should consider their innovative profile and/or the development of an innovative profile among current managers. Furthermore, when allocating employees to work units according to their professional profile, ensure that innovation is part of their employee profile. In addition, the training unit should create Learning Trails in Digital Competence, where the employee aims to carry out a sequence of training activities to acquire knowledge in digital competence and learn how to operate digital platforms. Such measures aim to mitigate potential digital gaps and strengthen innovative behaviors. In this line, digital competence enables the improvement of public service employees through training for digital technologies. Innovative employees can learn to reformulate work situations from an ecosystem perspective of innovation, bringing intentions to their interpersonal relationships.

Third, educational management in Federal Institutions of Higher Education should establish an effective digitization strategy. For this purpose, people management and information technology units need to provide an appropriate digital infrastructure in order to provide all necessary support for the use of information technologies.

By analyzing different dimensions of variables, this article can help educational managers in future developments and alert them in the process of building an information system. Educational institutions should pay attention to the value of innovation and technology content, rather than putting all the emphasis on traditional teaching and neglecting the essence of knowledge dissemination. On the other hand, during the innovation process, managers should encourage employees to express themselves and help them realize their innovative ideas, rather than just stopping at generating ideas. Moreover, this work can be used as a basis for decision making for educational managers to intervene in innovation and employee performance, to provide a theoretical basis for subsequent intervention strategies, and to help the public service to improve employees' work outcomes more effectively from a management perspective.

5.4. Limitations and Future Research Directions

In this context, Brazilian public management should consider digital competence as a fundamental factor to generate innovative work behavior among public employees. However, investment in technology does not guarantee innovative behavior, because other factors at the individual, group, and organizational level can influence and determine individual innovation to facilitate or inhibit its initiative in organizational contexts. In addition, creating an environment and promoting innovative work behavior is a challenge for public institutions due to their very highly bureaucratic and traditional nature.

Despite the validation and confirmation of the hypotheses investigated, this study was limited to investigating the influence of digital competence on innovative work behaviors of public service employees in the context of a Brazilian Federal Higher Education Institution. Additional work is suggested to improve understanding and test the influence of new factors such as mediators/moderators in this relationship, including support from management and the work team, self-esteem and self-efficacy (Attiq et al. 2017), knowledge sharing (Afsar et al. 2019; Bos-Nehles and Veenendaal 2019), and training and development and reward and autonomy at work (Bos-Nehles et al. 2017).

Federal Institutions of Higher Education are important sites for the changing developments of innovative behavior at work. First, managers need to adapt to technological changes in education. In this paper, we do not observe how much educational strategy impacts the innovative behavior of employees. As a result of planning, it is expected that results will be achieved to boost educational performance. Thus, it is worth broadening the reflection on the decision to invest in projects of excellence that do not leave the research stage. On the other hand, more engaged leaders and employees contributing to the education institution's strategies promote the building of an innovative culture and a thriving IWB. In addition, systematic solutions for the implementation of result for sustainable innovation with the creation of a culture of innovation in education, where experimentation and risk are encouraged and ideas are valued, must be encouraged.

6. Conclusions

The digital competence of a university could have a significant impact on the innovative behavior of its employees, made up of students, professors, researchers, and managers. Well-developed Digital Infrastructure, Digital Integration, and Digital Management could provide an environment conducive to creativity and experimentation, allowing the realization of innovative projects that could result in significant benefits to society. With a robust technological infrastructure, students and researchers could work with advanced tools that assist them in exploring innovative ideas and quickly test them. These technologies could include computer simulations, data modeling, and other tools for real-time experimentation. In addition, a solid digital competence could support external collaboration among university members, both locally and globally. High-speed networks enable real-time communication with other researchers on a global scale, which can lead to new ideas and innovative solutions.

One of the strategies for a Federal Higher Education Institution (IFES) to become innovative is by obtaining innovative behaviors from public officials. Given this context, IFES must identify and improve the infrastructure, integration, and digital management that influence the innovative work behavior of these employees. This study identified new elements that influence innovative work behavior. The results support the theoretical support of the conceptual structure of this research and indicate that the adoption of digital competence stimulates innovative work behaviors in public service employees. Therefore, this research provides a new perspective to IFES managers in Brazil to identify how the factors of digital competence can influence behaviors aimed at innovation at work.

Thus, IFES in Brazil, which intend to stimulate innovative behaviors in the organizational environment, need to understand the process of achieving digital competence in order to consider all its factors (infrastructure, integration, and digital management), which are interdependent and are linked to each other in a continuous cycle. First, an IT infrastructure should be made available as part of the organizational strategy in order to meet the needs of the organization. Then, effective integration of digital technologies to work processes must be promoted, and possible bottlenecks to integration must be identified and mapped in order to propose solutions. Finally, the knowledge related to the use of digital technologies in the organization must be managed in order to obtain the necessary knowledge to operate the systems and be always digitally up to date.

19 of 25

development meet adequate levels of reliability and validity (Saltos-Rivas et al. 2021). The assessment of the real level of digital competence is a research gap indicated by Zhao et al. (2021), implying in the need for new investigations, especially considering the matter of validation and reliability previously indicated. In complement to the public higher education perspective, future research can also apply a similar investigation to private higher education institutions.

Author Contributions: Conceptualization, L.P.d.C. and T.P.; methodology, L.P.d.C. and T.P.; software, L.P.d.C. and T.P.; validation, L.P.d.C. and T.P.; formal analysis, L.P.d.C. and T.P.; investigation, L.P.d.C. and T.P.; resources, L.P.d.C. and T.P.; data curation, L.P.d.C. and T.P.; writing—original draft preparation, L.P.d.C. and T.P.; writing—review and editing, L.P.d.C., T.P., C.C.R., F.d.A.R., V.D.H.d.C. and T.C.C.N.; visualization, L.P.d.C., T.P., C.C.R., F.d.A.R., V.D.H.d.C. and T.C.C.N.; supervision, L.P.d.C., T.P., C.C.R., F.d.A.R., V.D.H.d.C. and T.C.C.N.; project administration, L.P.d.C., T.P., C.C.R., F.d.A.R., V.D.H.d.C. and T.C.C.N.; P.d.A.R., V.D.H.d.C. and T.C.C.N. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: The data presented in this study are available upon request from the corresponding author.

Acknowledgments: This research was partially supported by the Federal University of PARA and Dean of Research and Graduate Studies—PROPESP—The authors would like to acknowledge the National Council for the Improvement of Higher Education (CAPES) a foundation affiliated with the Ministry of Education in Brazil.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A. Questionnaire

Digital Infrastructure

Refers to ownership and use of digital-related infrastructure by the institution.

DI1. You use different digital technologies (emails; applications such as: whatsapp, telegram, instagram and facebook; website and information systems of the institution, such as: SIGAA, SIPAC, SIGRH, SAGITTA and/or SINPEG) to improve communication with the community university.

- 1. I do not use digital technologies for communication.
- 2. I use basic digital technologies for communication, for example: e-mail.
- 3. Sometimes I combine different digital technologies for communication.
- 4. Occasionally, I select, adjust and combine different digital solutions to communicate effectively.
- 5. I often reflect, discuss and proactively develop my communication strategies, in order to adopt the technologies that best suit my reality.

DI2. You use digital technologies (emails; messaging apps such as: whatsapp and telegram; videoconferencing platforms, such as: google meet, zoom and teams; cloud file storage platforms, such as: google drive and onedrive; management information systems, such as: SIGAA, SIGRH, SIPAC, SAGITTA and/or SINPEG) to work with colleagues inside and outside your institution.

- 1. It is not possible in my work context.
- 2. I rarely have the opportunity to use digital technologies to work with colleagues.
- 3. I sometimes use digital technologies to work with colleagues inside and outside my institution.
- 4. I occasionally use digital technologies to work with colleagues inside and outside my institution.
- 5. I frequently use digital technologies to work with colleagues inside and outside my institution.

DI3. You carefully consider how, when and why to use digital technologies in your unit's work processes to ensure they add value.

- 1. I do not use digital technologies in my work processes.
- 2. I rarely make basic use of available digital technologies.
- 3. Sometimes I use some digital resources and tools to solve specific problems in my work processes.
- 4. I occasionally use digital technologies to improve the execution of my work processes.
- 5. I often use various digital technologies to implement innovative strategies in my work processes.

DI4. In your perception, the digital infrastructure of your institution's information systems (SIGAA, SIGRH, SIPAC, SAGITTA and/or SINPEG) is capable of meeting the needs of work processes in your unit.

- 1. I have not yet had the opportunity to work with the information systems at my institution.
- 2. My institution's information systems do not meet the needs of my work processes.
- 3. My institution's information systems barely meet the needs of my work processes.
- 4. My institution's information systems satisfy, to a large extent, the needs of my work processes.
- 5. My institution's information systems fully satisfy the needs of my work processes.

DI5. You have adequate digital knowledge and skills to use your institution's digital infrastructure (SIGAA, SIGRH, SIPAC, SAGITTA and/or SINPEG).

- 1. This is a new area that I haven't considered yet.
- 2. Not yet, but I'm definitely interested in acquiring these skills.
- 3. I have insufficient digital knowledge and skills to use my institution's digital infrastructure.
- 4. I have reasonable digital knowledge and skills to use my institution's digital infrastructure.
- 5. I have excellent digital knowledge and skills to utilize my institution's digital infrastructure.

Digital Integration

Refers to the institution's integration of various digital-related strategies, technologies and knowledge resources.

DIT1. In your perception, his Unit's Development Plan (UDP) is integrated with the institution's digital strategy, in order to contemplate the use of digital technologies (SIGAA, SIGRH, SIPAC, SAGITTA and/or SINPEG).

- 1. My work unit does not intend to integrate the institution's digital strategy.
- 2. Digital strategy is a new area that has not yet been considered in UDP of my unit.
- 3. Not yet, but my work unit is definitely interested.
- 4. Not yet, but my work unit is reformulating its UDP to include the digital strategy.
- 5. Yes, the UDP of my unit is integrated into the institution's digital strategy.

DIT2. In your perception, your work unit integrates digital technologies in all its subunits, through the use of the institution's information systems platforms (SIGAA, SIGRH, SIPAC, SAGITTA and/or SINPEG).

- 1. Digital integration is not possible in my work unit.
- 2. My work unit does not intend to integrate digital technologies into its subunits.
- 3. My work unit integrates digital technologies in some subunits.
- 4. My work unit integrates digital technologies into most of its subunits.
- 5. My work unit integrates digital technologies into all its subunits.

DIT3. In your perception, your institution's information systems (SIGAA, SIGRH, SIPAC, SAGITTA and/or SINPEG) enable the integration of data between different subunits.

- 1. My institution's information systems do not allow the integration of data between different subunits.
- 2. My institution's information systems allow insufficient integration of data between different subunits.
- 3. My institution's information systems allow a reasonable integration of data between several subunits.
- 4. My institution's information systems allow good integration of data between different subunits.
- 5. My institution's information systems allow for excellent integration of data across multiple subunits.

DIT4. You are able to freely share and use data from other subunits through the institution's information systems (SIGAA, SIGRH, SIPAC, SAGITTA and/or SINPEG).

- 1. I do not use the institution's information systems to share and use data.
- 2. I am not able to freely share and use data from other subunits through the institution's information systems.
- I am reasonably able to basically use the institution's information systems to freely share and use data from other subunits.
 Optimally, I am able to search the institution's different information systems for data from other subunits to share and use
- freely.
- Excellently, I am able to combine different information systems of the institution to freely share and use data from other subunits.

DIT5. You consider and address potential practical or technical difficulties in making digital services available to the university community.

For example: "interoperability and conversion issues, lack of digital skills".

- 1. I don't create digital services.
- 2. I do not consider potential difficulties, as students, teachers and/or technicians have no problems using digital technology.
- 3. I basically adapt digital services to minimize difficulties.
- 4. I discuss possible obstacles with students, teachers and/or technicians and outline solutions to potential difficulties.
- 5. I address a variety of possible difficulties, for example, adapt services, discuss solutions and provide alternative paths.

Digital Management

Refers to the acquisition of skills and experience related to digital by the management of the institution.

DM1. You actively develop your skills in digital technologies.

- 1. I don't have time to improve my skills in digital technologies.
- 2. I rarely improve my skills through reflection and experimentation.
- 3. I sometimes use a variety of digital resources to develop my skills at work.
- 4. Occasionally, I discuss with colleagues how to use digital technologies to innovate and improve work processes.
- 5. I often develop, and help colleagues develop, strategies to improve the use of digital technologies at work.

DM2. You participate in online training when you have the opportunity. (For example: online courses, webinars, virtual conferences).

- 1. This is a new area that I haven't considered yet.
- 2. Not yet, but I'm definitely interested.
- 3. I attended online training once or twice.
- 4. I occasionally participate in some online training.
- 5. I often participate in all kinds of online training.

DM3. From the digital training offered by the institution, you apply the skills and knowledge of digital systems (SIGAA, SIGRH, SIPAC, SAGITTA and/or SINPEG).

- 1. This is a new area that I haven't considered yet.
- 2. I still can't apply my knowledge of digital systems.
- 3. Apply basic knowledge of digital systems.
- 4. I apply intermediate knowledge of digital systems.
- 5. I apply complex knowledge of digital systems.

DM4.You use the institution's digital systems (SIGAA, SIGRH, SIPAC, SAGITTA and/or SINPEG) to solve internal and external demands related to your work processes.

- 1. I don't use digital systems.
- 2. I don't need digital systems to solve internal and external demands.
- 3. Sometimes, I use digital systems to solve internal and external demands.
- 4. Occasionally, I use various digital systems to solve internal and external demands.
- 5. I often integrate different digital systems to solve internal and external demands.

DM5. You plan, document and monitor your work processes using digital technologies. For example: use of videoconferencing platforms (google meet, zoom, teams), cloud file storage platforms (google drive, onedrive), management information systems such as: SIGAA, SIGRH, SIPAC, SAGITTA and/or SINPEG.

- 1. It is not possible in my work context.
- 2. I do not adopt digital technologies in my work context.
- 3. Sometimes I use, for example, a cloud file storage platform to store documentation.
- 4. I occasionally use digital technologies to plan, document and monitor my work processes.
- 5. I often integrate different digital technologies to plan, document and monitor my work processes.

Innovative Work Behavior

It refers to behaviors of "individuals" directed towards the intentional initiation and introduction of new and useful ideas, processes, products or procedures.

IWB1. You look for opportunities to improve processes or services provided to the university community.

- 1. I do not look for opportunities to improve my work processes.
- 2. I rarely look for opportunities to collaborate on improving my work processes.
- 3. Sometimes I look for opportunities to improve my work processes.
- 4. I occasionally look for opportunities to improve my work processes.
- 5. I often look for opportunities to improve my work processes.

IWB2. You generate ideas or solutions to improve services provided to the university community.

- 1. I do not generate ideas or solutions to improve my work processes.
- 2. I rarely generate ideas and/or solutions to improve my work processes.
- 3. Sometimes I generate ideas and/or solutions to improve my work processes.
- 4. Occasionally, I generate ideas and/or solutions to improve my work processes.
- 5. I often generate ideas and/or solutions to improve my work processes.

IWB3. You participate in debates and meetings in order to contribute new ideas to your Unit Development Plan (PDU).

- 1. I have no interest in contributing new ideas to my unit's PDU.
- 2. Not yet, but I am interested in attending meetings to contribute new ideas to my unit's PDU.
- 3. I sometimes attend meetings to contribute innovative ideas to my unit's PDU.
- 4. I occasionally participate in meetings in order to contribute innovative ideas to my unit's PDU.
- 5. I often participate in meetings to contribute innovative ideas to my unit's PDU.

IWB4. You encourage and encourage co-workers in the implementation of new ideas in favor of improving service to the university community.

- 1. I have no interest in encouraging coworkers to implement new ideas.
- 2. I rarely encourage my co-workers to implement new ideas.
- 3. Sometimes, I exchange experiences with colleagues in order to encourage them to implement new ideas to improve work processes.
- 4. Occasionally, I work with colleagues to propose improvements in work processes.
- 5. I frequently exchange ideas with colleagues in order to encourage them to add improvements to work processes.

IWB5. You adopt the institution's digital systems (SIGAA, SIGRH, SIPAC, SAGITTA and/or SINPEG) in your work processes, with the aim of improving the fulfillment of demands from the university community.

- 1. It is not possible in my work context.
- 2. I do not use the institution's digital systems in my work processes.
- 3. I adopt some of the institution's digital systems in my work processes.
- 4. I adopt several digital systems of the institution in my work processes.
- 5. I adopt all the institution's digital systems that can be used in my work processes.

Respondent Profile Identification	Answers	
Sex	() Female ()male	
Age Range	() 18–30 years old () 31–40 years old () 41–50 years old () Over 51 years old	
Work Unit	 () Pro-Rectory of Undergraduate Education (PROEG) () Pro-Rectory of Research and Graduate Studies (PROPESP) () Pro-Rectory of Extension (PROEX) () Pro-Rectory of International Relations (PROINTER) () Pro-Rectory of Administration (PROAD) () Pro-Rectory of Personnel Development and Management (PROGEP) () Pro-Rectory of Planning and Institutional Development (PROPLAN) () City hall () Information and Communication Technology Center (CTIC) () Central Archive () Technological Innovation Agency (UNIVERSITEC) () Institute of Geosciences (IG) () Other	
Education Level	 () Complete Elementary School () Complete Higher Education () Graduate at the Master's level () Post-Doctorate () Complete High School () Complete High School () Graduate at the Specialization/MBA level () Graduate at the Doctorate level 	
knowledge Area	() Administration () Archivology () Librarianship () Accounting Sciences () Economic Sciences () Engineering () Information Technology () Other	

References

- Afsar, Bilal, and Waheed Ali Umrani. 2020. Transformational Leadership and Innovative Work Behavior: The Role of Motivation to Learn, Task Complexity and Innovation Climate. *European Journal of Innovation Management* 23: 402–28. [CrossRef]
- Afsar, Bilal, Mariam Masood, and Waheed Ali Umrani. 2019. The Role of Job Crafting and Knowledge Sharing on the Effect of Transformational Leadership on Innovative Work Behavior. *Personnel Review* 48: 1186–208. [CrossRef]
- Afsar, Bilal, Yuosre Badir, and Bilal Saeed. 2014. Transformational Leadership and Innovative Work Behavior. *Industrial Management* and Data Systems 114: 1270–300. [CrossRef]
- Afsar, Bilal, Yuosre Badir, and Muhammad Muddassar Khan. 2015. Person-Job Fit, Person-Organization Fit and Innovative Work Behavior: The Mediating Role of Innovation Trust. *Journal of High Technology Management Research* 26: 105–16. [CrossRef]
- Ala-mutka, Kirsti. 2011. Mapping Digital Competence: Towards a Conceptual Understanding. JRC European Commission 2011: 7–60. [CrossRef]
- Amhag, Lisbeth, Lisa Hellström, and Martin Stigmar. 2019. Teacher Educators' Use of Digital Tools and Needs for Digital Competence in Higher Education. Journal of Digital Learning in Teacher Education 35: 203–20. [CrossRef]
- Andersson, Annika, Karin Hedström, Steffi Siegert, and Carl Johan Sommar. 2021. Teachers Falling off the Cliff Affordances and Constraints of Social Media in School. Paper presented at the Annual Hawaii International Conference on System Sciences, Maui, HI, USA, January 7–10; vol. 2020, pp. 2995–3004. [CrossRef]
- Antonucci, Yvonne Lederer, Annetta Fortune, and Mathias Kirchmer. 2021. An Examination of Associations between Business Process Management Capabilities and the Benefits of Digitalization: All Capabilities Are Not Equal. *Business Process Management Journal* 27: 124–44. [CrossRef]
- Atitumpong, Aungkhana, and Yuosre F. Badir. 2018. Leader-Member Exchange, Learning Orientation, and Innovative Work Behavior. Journal of Workplace Learning 30: 32–47. [CrossRef]
- Attali, Yigal, and Meirav Arieli-attali. 2015. Computers & Education Gami Fi Cation in Assessment: Do Points Affect Test Performance? Computers & Education 83: 57–63. [CrossRef]
- Attiq, Saman, Sidra Wahid, Nimra Javaid, Maria Kanwal, and Hassan J. Shah. 2017. The Impact of Employees' Core Self-Evaluation Personality Trait, Management Support, Co-Worker Support on Job Satisfaction, and Innovative Work Behaviour. Pakistan Journal of Psychological Research 32: 247–71.
- Austin, Robert D., Lee Devin, and Erin E. Sullivan. 2012. Accidental Innovation: Supporting Valuable Unpredictability in the Creative Process. Organization Science 23: 1505–22. [CrossRef]
- Ayoub, Alaa Eldin Abdel Hamid, Soud Mohammad Almahamid, and Luma F. Al Salah. 2023. Innovative work behavior scale: Development and validation of psychometric properties in higher education in the GCC countries. *European Journal of Innovation Management* 26: 119–33. [CrossRef]
- Baig, Maria Ijaz, Liyana Shuib, and Elaheh Yadegaridehkordi. 2020. Big Data in Education: A State of the Art, Limitations, and Future Research Directions. International Journal of Educational Technology in Higher Education 17: 1–23. [CrossRef]
- Basantes-Andrade, Andrea, Marcos Cabezas-González, and Sonia Casillas-Martín. 2020. Digital Competences Relationship between Gender and Generation of University Professors. International Journal on Advanced Science, Engineering and Information Technology 10: 205–11. [CrossRef]
- Basilotta-Gómez-Pablos, Verónica, María Matarranz, Luis Alberto Casado-Aranda, and Ana Otto. 2022. Teachers' Digital Competencies in Higher Education: A Systematic Literature Review. *International Journal of Educational Technology in Higher Education* 19: 1–16. [CrossRef]
- Battistelli, Adalgisa, Francesco Montani, and Carlo Odoardi. 2013. The Impact of Feedback from Job and Task Autonomy in the Relationship between Dispositional Resistance to Change and Innovative Work Behaviour. *European Journal of Work and Organizational Psychology* 22: 26–41. [CrossRef]
- Bharadwaj, Anandhi, Omar A. El Sawy, Paul A. Pavlou, and N. Venkat Venkatraman. 2013. Digital Business Strategy: Toward a Next Generation of Insights. MIS Quarterly 37: 471–82. [CrossRef]
- Bos-Nehles, Anna C., and Andre A. R. Veenendaal. 2019. Perceptions of HR Practices and Innovative Work Behavior: The Moderating Effect of an Innovative Climate. *International Journal of Human Resource Management* 30: 2661–83. [CrossRef]
- Bos-Nehles, Anna C., Maarten Renkema, and Maike Janssen. 2017. HRM and Innovative Work Behaviour: A Systematic Literature Review. Personnel Review 46: 1228–53. [CrossRef]
- Cabero-Almenara, Julio, Julio Barroso-Osuna, Antonio Palacios-Rodríguez, and Carmen Llorente-Cejudo. 2020. Digital Competency Frames for University Teachers: Evaluation through the Expert Competence Coefficient. *Revista Electronica Interuniversitaria de Formacion Del Profesorado* 23: 1–18.
- Caena, Francesca, and Christine Redecker. 2019. Aligning Teacher Competence Frameworks to 21st Century Challenges: The Case for the European Digital Competence Framework for Educators (DIGCOMPEDU). *European Journal of Education* 54: 356–69. [CrossRef]
- Castioni, Remi, Adriana Almeida Sales de Melo, Paulo Meyer Nascimento, and Daniela Lima Ramos. 2021. Brazilian Federal Universities in the COVID-19 Pandemic: Student Internet Access and Emergency Remote Teaching. *Ensaio: Avaliação e Políticas Públicas em Educação* 29: 399–419. [CrossRef]
- Ciriza-Mendívil, Carlos D., Ana Mendioroz Lacambra, and José M. Hernández de la Cruz. 2022. Technological Pedagogical Content Knowledge: Implementation of a Didactic Proposal for Preservice History Teachers. *Frontiers in Education* 7: 852801. [CrossRef]

Cohen, Jacob. 1988. Statistical Power Analysis for the Behavioral Sciences, 2nd ed. Hillsdale: Erlbaum.

- Dahiya, Rinki, and Juhi Raghuvanshi. 2021. Validation of Innovative Work Behaviour Scale: Indian Apparel Manufacturing Sector. *Asia Pacific Management Review* 27: 120–36. [CrossRef]
- Darazha, Issabayeva, Rakhimzhanova Lyazzat, Abdigapbarova Ulzharkyn, Zhiyenbayeva Saira, and Zhumartov Manat. 2021. Digital Competence of a Teacher in a Pandemic. Paper presented at the 2021 9th International Conference on Information and Education Technology, ICIET 2021, Okayama, Japan, March 27–29; pp. 324–28. [CrossRef]
- De Jong, Jeroen, and Deanne Den Hartog. 2010. Measuring Innovative Work Behaviour. Creativity and Innovation Management 19: 23–36. [CrossRef]
- De Spiegelaere, Stan, Guy Van Gyes, and Geert Van Hootegem. 2014. The Innovative Work Behaviour Concept: Definition and Orientation. *Gedrag En Organisatie* 27: 139–56. [CrossRef]
- Dias-Trindade, Sara, and António Gomes Ferreira. 2020. Digital Teaching Skills: Digcompedu Checkin as an Evolution Process from Literacy to Digital Fluency. *Icono14* 18: 162–87. [CrossRef]
- Dobson, Rosie, Rushaina Variava, Meihana Douglas, and Lisa M. Reynolds. 2022. Digital Competency of Psychologists in Aotearoa New Zealand: A Cross-Sectional Survey. *Frontiers in Digital Health* 4: 951366. [CrossRef]
- Dospinescu, Octavian, and Nicoleta Dospinescu. 2020. Perception Over E-Learning Tools in Higher Education: Comparative Study Romania and Moldova. Paper presented at the 19th International Conference on Informatics in Economy Education, Research and Business Technologies, Vienna, Austria, September 21–23; vol. 2020, pp. 59–64. [CrossRef]
- Fadli, Moh, Diah Pawestri Maharani, and Airin Liemanto. 2020. A New Paradigm of Regulation for Massive Open Online Courses (MOOCs) in Higher Education in Indonesia: From Disruptive Innovation to Sustaining Innovation. International Journal of Innovation, Creativity and Change 11: 442–57.
- Fornell, Claes, and David F. Larcker. 1981. Evaluating Structural Equation Models with Unobservable Variables and Measurement Error. *Journal of Marketing Research* 18: 39–50. [CrossRef]
- Galanti, Teresa, Clara De Vincenzi, Ilaria Buonomo, and Paula Benevene. 2023. Digital Transformation: Inevitable Change or Sizable Opportunity? The Strategic Role of HR Management in Industry 4.0. *Administrative Sciences* 13: 30. [CrossRef]
- Gansser, Oliver, and Carsten D Schultz. 2020. Perception of Digitalization across a Typology of Consumer Behavior: An Abstract BT. In Enlightened Marketing in Challenging Times. Edited by Felipe Pantoja, Shuang Wu and Nina Krey. Cham: Springer International Publishing, pp. 305–6.
- Gerbaulet, Damian, and Oliver Korn. 2018. Do Zenware Applications Reduce the Digital Distraction of Knowledge Workers? A Qualitative Study Based on Expert Interviews. *Advances in Intelligent Systems and Computing* 588: 115–26. [CrossRef]
- Gleason, Benjamin, and Stefania Manca. 2020. Curriculum and Instruction: Pedagogical Approaches to Teaching and Learning with Twitter in Higher Education. *On the Horizon* 28: 1–8. [CrossRef]
- Guillén-Gámez, Francisco D., and María J. Mayorga-Fernández. 2020. Prediction of Factors That Affect the Knowledge and Use Higher Education Professors from Spain Make of Ict Resources to Teach, Evaluate and Research: A Study with Research Methods in Educational Technology. *Education Sciences* 10: 276. [CrossRef]
- Hair, Joseph F., G. Tomas M. Hult, Christian Ringle, and Marko Sarstedt. 2017. A Primer on Partial Least Squares Structural Equation Modeling (PLS-SEM), 2nd ed. Los Angeles: SAGE.
- Hashim, Mohamed Ashmel Mohamed, Issam Tlemsani, Robin Matthews, Rachel Mason-Jones, and Vera Ndrecaj. 2022. Emergent Strategy in Higher Education: Postmodern Digital and the Future? *Administrative Sciences* 12: 196. [CrossRef]
- Henseler, Jörg, Christian M. Ringle, and Marko Sarstedt. 2015. A New Criterion for Assessing Discriminant Validity in Variance-based Structural Equation Modeling. *Journal of the Academy of Marketing Science* 43: 115–35. [CrossRef]
- Henseler, Jörg, Christian M. Ringle, and Rudolf R. Sinkovics. 2009. The Use of Partial Least Squares Path Modeling in International Marketing. *Advances in International Marketing* 20: 277–319. [CrossRef]
- Hess, Thomas, Alexander Benlian, Christian Matt, and Florian Wiesböck. 2016. How German Media Companies Defined Their Digital Transformation Strategies. *MIS Quarterly Executive* 15: 103–19.
- Hoti, Irida, Blerta Dragusha, and Valentina Ndou. 2022. Online Teaching during the COVID-19 Pandemic: A Case Study of Albania. *Administrative Sciences* 12: 116. [CrossRef]
- Kindermann, Bastian, Sebastian Beutel, Gonzalo Garcia de Lomana, Steffen Strese, David Bendig, and Malte Brettel. 2021. Digital Orientation: Conceptualization and Operationalization of a New Strategic Orientation. *European Management Journal* 39: 645–57. [CrossRef]
- Kock, Ned. 2015. One-tailed or two-tailed P values in PLS-SEM? International Journal of e-Collaboration 11: 1–7. [CrossRef]
- Li, Jun, Michael Merenda, and A. R. Venkatachalam. 2009. Business Process Digitalization and New Product Development: An Empirical Study of Small and Medium-Sized Manufacturers. *International Journal of E-Business Resarch* 5: 49–64. [CrossRef]
- Lindberg, Monika, Maria Ranner, Eva Månsson-Lexell, Lars Jacobsson, and Maria Larsson-Lund. 2021. Work and Everyday Life in a Digitalized Time: Experiences of People with Subjective Cognitive Difficulties Related to Neurological Disorders. *PLoS ONE* 16: e0260013. [CrossRef] [PubMed]
- Liu, Sen, Felix T. S. Chanb, Junai Yang, and Ben Niu. 2018. Understanding the effect of cloud computing on organizational agility: An empirical examination. *International Journal of Information Management* 43: 98–111.
- Maria Stock, Ruth, Nicolas A. Zacharias, and Armin Schnellbaecher. 2017. How Do Strategy and Leadership Styles Jointly Affect Co-Development and Its Innovation Outcomes? *Journal of Product Innovation Management* 34: 201–22. [CrossRef]

- Nascimento, João Carlos Hipólito Bernardes do, and Marcelo Alvaro da Silva Macedo. 2016. Modeling of Structural Equations with Partial Least Squares: An Example of the Application of SmartPLS in Accounting Research. *Revista de Educação e Pesquisa Em Contabilidade (REPeC)* 10: 289–313. [CrossRef]
- Nikou, Shahrokh, and Milla Aavakare. 2021. An Assessment of the Interplay between Literacy and Digital Technology in Higher Education. *Education and Information Technologies* 26: 3893–915. [CrossRef]
- Paschou, Theoni, Mario Rapaccini, Federico Adrodegari, and Nicola Saccani. 2020. Digital Servitization in Manufacturing: A Systematic Literature Review and Research Agenda. *Industrial Marketing Management* 89: 278–92. [CrossRef]
- Pesha, Anastasia. 2022. The Development of Digital Competencies and Digital Literacy in the 21st Century: A Survey of Studies. *Education and Self Development* 17: 201–20. [CrossRef]
- Quinton, Sarah, Ana Isabel Canhoto, Sebastian Molinillo, and Rebecca Pera. 2018. Conceptualising a Digital Orientation: Antecedents of Supporting SME Performance in the Digital Economy. *Journal of Strategic Marketing* 26: 427–39. [CrossRef]
- Ramos, Daniela Karine, Bruna Santana Anastácio, Gleice Assunção da Silva, Leila Urioste Rosso, and João Mattar. 2023. Burnout Syndrome in Different Teaching Levels during the COVID-19 Pandemic in Brazil. *BMC Public Health* 23: 1–11. [CrossRef]
- Ravichandran, Thiagarajan. 2018. Exploring the Relationships between IT Competence, Innovation Capacity and Organizational Agility. *Journal of Strategic Information Systems* 27: 22–42. [CrossRef]
- Redecker, Christine. 2017. European Framework for the Digital Competence of Educators: DigCompEdu. In *Joint Research Centre (JRC)* Science for Policy Report. Edited by Yves Punie. Luxembourg: Publications Office of the European Union. [CrossRef]
- Ringle, Christian M., Dirceu Da Silva, and Diógenes De Souza Bido. 2014. Modelagem de Equações Estruturais Com Utilização Do Smartpls. *Revista Brasileira de Marketing* 13: 56–73. [CrossRef]
- Rivero, Pilar, and Lorenzo Mur. 2015. Learning Social Sciences with web 2.0. *Íber Didáctica de Las Ciencias Sociales, Geografía e Historia* 80: 30–37.
- Saeed, Bilal Bin. 2019. Leader-Member Exchange and Innovative Work Behavior. *European Journal of Innovation Management* 22: 105–24. [CrossRef]
- Saltos-Rivas, Rafael, Pavel Novoa-Hernández, and Rocío Serrano Rodríguez. 2021. On the Quality of Quantitative Instruments to Measure Digital Competence in Higher Education: A Systematic Mapping Study. *PLoS ONE* 16: e0257344. [CrossRef]
- Sawy, Omar A. El, Henrik Amsinck, Pernille Kræmmergaard, and Anders Lerbech Vinther. 2016. How LEGO Built the Foundations and Enterprise Capabilities for Digital Leadership. *MIS Quarterly Executive* 15: 141–66.
- Sengupta, Debashish, and Dwa Al-Khalifa. 2022. Pandemic Imposed Remote Work Arrangements and Resultant Work-Life Integration, Future of Work and Role of Leaders—A Qualitative Study of Indian Millennial Workers. *Administrative Sciences* 12: 162. [CrossRef]
- Sengupta, Santoshi, Badri Bajaj, Aishwarya Singh, Swati Sharma, Parth Patel, and Verma Prikshat. 2023. Innovative work behavior driving Indian startups go global—The role of authentic leadership and readiness for change. *Journal of Organizational Change Management* 36: 162–79. [CrossRef]
- Sia, Siew Kien, Christina Soh, and Peter Weill. 2016. How DBS Bank pursued a digital business strategy. *MIS Quarterly Executive* 15: 105–21.
- Tejada Fernández, José, and Katia V. Pozos Pérez. 2018. New scenarios and trainers' digital competencies: Towards the professiona ization of teaching with ICT. *Profesorado, Revista de Currículum y Formación Del Profesorado* 22: 25–51. [CrossRef]
- Van de Wetering, Rogier, Sherah Kurnia, and Svyatoslav Kotusev. 2021. The Role of Enterprise Architecture for Digital Transformations. Sustainability 13: 2237. [CrossRef]
- Verhoef, Peter C., Thijs Broekhuizen, Yakov Bart, Abhi Bhattacharya, John Qi Dong, Nicolai Fabian, and Michael Haenlein. 2021. Digital Transformation: A Multidisciplinary Reflection and Research Agenda. *Journal of Business Research* 122: 889–901. [CrossRef]
- Weking, Jörg, Michael Mandalenakis, Andreas Hein, Sebastian Hermes, Markus Böhm, and Helmut Krcmar. 2020. The Impact of Blockchain Technology on Business Models—A Taxonomy and Archetypal Patterns. *Electronic Markets* 30: 285–305. [CrossRef]
- Yu, Jiatong, and Taesoo Moon. 2021. Impact of Digital Strategic Orientation on Organization Performance through Digital Transformation Capability. ICIC Express Letters, Part B: Applications 12: 847–56. [CrossRef]
- Yu, Jinmeng, Jinlan Liu, Sheng Lin, and Xianglan Chi. 2023. The relationship between stressors and Chinese employees' innovative work behavior: The role of task crafting and psychological detachment. *European Journal of Innovation Management. ahead-of-print*. [CrossRef]
- Zhao, Yu, Ana María Pinto Llorente, and María Cruz Sánchez Gómez. 2021. Digital Competence in Higher Education Research: A Systematic Literature Review. *Computers and Education* 168: 104212. [CrossRef] [PubMed]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.